Author's response to reviews

Title: Bilateral brachial plexus injury following acute carbon monoxide poisoning

Authors:

Mounia Rahmani (mouniarahmani4@gmail.com)
Halima Belaidi (halima-belaidi@hotmail.fr)
Maria Benabdeljjil (benab.maria@yahoo.fr)
Wafa Bouchhab (bo.wafa.md@gmail.com)
Nadia El Jazouli (nadia.eljazouli@gmail.com)
Asmae El Brini (dr.brini.neuro@hotmail.fr)
Saadia Aidi (aidi_sadia@yahoo.fr)
Reda M Ouazzani (rida@menara.ma)
Mustapha El Alaoui Faris (elalaouifarism@yahoo.fr)

Version: 2 Date: 3 January 2013

Author's response to reviews: see over
This article is very interesting because it describes the first case of bilateral brachial plexus injury following acute monoxide (CO) poisoning. Otherwise it reviews all published reports of peripheral neuropathy induced by CO intoxication. It shows hence that in addition to major neuropsychological complications, CO can lead to neuropathy.